Author's response to reviews

Title: A Comparison of Online versus On-site Training in Health Research
Methodology: A Randomized Study

Authors:

- Rakesh Aggarwal (aggarwal.ra@gmail.com)
- Nikhil Gupte (nikhil_jhumit@yahoo.com)
- Nancy Kass (nkass@jhsph.edu)
- Holly Taylor (htaylor@jhsph.edu)
- Joseph Ali (jali@jhsph.edu)
- Anant Bhan (anantbhan@gmail.com)
- Amita Aggarwal (aa.amita@gmail.com)
- Stephen D Sisson (ssisson@jhmi.edu)
- Sukon Kanchanaraksa (skanchan@jhsph.edu)
- Jane McKenzie-White (jmw@jhmi.edu)
- John McGready (jmcgread@jhsph.edu)
- Paolo Miotti (miottip@od.nih.gov)
- Robert C Bollinger (rcb@jhmi.edu)

Version: 5 Date: 16 June 2011

Author's response to reviews:

Have added footnotes to Tables 3, 4, 6 and 7

(originally there were only 4 tables. The original table 3 and 4 were split into three each, i.e. Tables 3-5 and 6-8, respectively. Due to error on my part, the footnote did not appear in some of the tables. I have now added that).